Author's response to reviews

Title: When does poor subjective financial position hurt the elderly? Testing the interaction with educational attainment using a national representative longitudinal survey

Authors:

Christy Pu (cypu@ym.edu.tw)
Nicole Huang (syhuang@ym.edu.tw)
Gao-Jun Tang (gjtang@ym.edu.tw)
Yiing-Jenq Chou (yjchou@ym.edu.tw)

Version: 6 Date: 9 March 2011

Author's response to reviews:

Dear Editorial office,

We are submitting the corrected proof. Thank you.

Regards,

Yiing-Jenq Chou